Author’s response to reviews

Title: National profile of foot orthotic provision in the United Kingdom, Part 1: Practitioners and scope of practice

Authors:

Christopher Nester (C.J.Nester@salford.ac.uk)
Andrea Graham (A.S.Graham@salford.ac.uk)
Ana Martinez (A.MartinezSantos@edu.salford.ac.uk)
Anita Williams (a.e.williams1@salford.ac.uk)
Jane McAdam (j.mcadam@salford.ac.uk)
Veronica Newton (v.a.newton@salford.ac.uk)

Version: 2 Date: 09 Jul 2017

Author’s response to reviews:

We have adjusted the title ensured the title in the JFAR online submission system is consistent with the paper title.

We have removed all use of capitals for common nouns as requested (in the main body of the text and tables).

On the issue of using numerals or words for numbers, and using both styles in one sentence, we have reviewed recent papers in JFAR and to conform with these we have made all numbers numerals.